



# Systematic Review IoT Adoption and Application for Smart Healthcare: A Systematic Review

Manal Al-rawashdeh <sup>1,\*</sup>, Pantea Keikhosrokiani <sup>1,\*</sup>, Bahari Belaton <sup>1</sup>, Moatsum Alawida <sup>1,2</sup> and Abdalwhab Zwiri <sup>3</sup>

- <sup>1</sup> School of Computer Sciences, Universiti Sains Malaysia, Penang 11800, Malaysia; bahari@usm.my (B.B.); moatsum.alawida@adu.ac.ae (M.A.)
- <sup>2</sup> Department of Computer Sciences, Abu Dhabi University, Abu Dhabi 59911, United Arab Emirate
- <sup>3</sup> School of Dental Sciences, Health Campus, Universiti Sains Malaysia, Kelantan 16150, Malaysia; abdalwhab@student.usm.my
- \* Correspondence: manal@student.usm.my (M.A.-r.); pantea@usm.my (P.K.)

Abstract: In general, the adoption of IoT applications among end users in healthcare is very low. Healthcare professionals present major challenges to the successful implementation of IoT for providing healthcare services. Many studies have offered important insights into IoT adoption in healthcare. Nevertheless, there is still a need to thoroughly review the effective factors of IoT adoption in a systematic manner. The purpose of this study is to accumulate existing knowledge about the factors that influence medical professionals to adopt IoT applications in the healthcare sector. This study reviews, compiles, analyzes, and systematically synthesizes the relevant data. This review employs both automatic and manual search methods to collect relevant studies from 2015 to 2021. A systematic search of the articles was carried out on nine major scientific databases: Google Scholar, Science Direct, Emerald, Wiley, PubMed, Springer, MDPI, IEEE, and Scopus. A total of 22 articles were selected as per the inclusion criteria. The findings show that TAM, TPB, TRA, and UTAUT theories are the most widely used adoption theories in these studies. Furthermore, the main perceived adoption factors of IoT applications in healthcare at the individual level are: social influence, attitude, and personal inattentiveness. The IoT adoption factors at the technology level are perceived usefulness, perceived ease of use, performance expectancy, and effort expectations. In addition, the main factor at the security level is perceived privacy risk. Furthermore, at the health level, the main factors are perceived severity and perceived health risk, respectively. Moreover, financial cost, and facilitating conditions are considered as the main factors at the environmental level. Physicians, patients, and health workers were among the participants who were involved in the included publications. Various types of IoT applications in existing studies are as follows: a wearable device, monitoring devices, rehabilitation devices, telehealth, behavior modification, smart city, and smart home. Most of the studies about IoT adoption were conducted in France and Pakistan in the year 2020. This systematic review identifies the essential factors that enable an understanding of the barriers and possibilities for healthcare providers to implement IoT applications. Finally, the expected influence of COVID-19 on IoT adoption in healthcare was evaluated in this study.

**Keywords:** IoT; IoMT; IoT adoption; systematic review; adoption theories; adoption factors; Internet of Things in healthcare; machine learning (ML); deep learning (DL)

# 1. Introduction

With increasing chronic illnesses, the number of patients in dire need of medical intervention is rising rapidly. This has invariably put pressure on healthcare services and delivery. Healthcare administrators, physicians, nurses, and other health professionals are facing increasing pressure to respond to the growing demands of both the public and the private sector on health-related matters. Additionally, the increasing cost of medical care has



Citation: Al-rawashdeh, M.; Keikhosrokiani, P.; Belaton, B.; Alawida, M.; Zwiri, A. IoT Adoption and Application for Smart Healthcare: A Systematic Review. *Sensors* 2022, 22, 5377. https:// doi.org/10.3390/s22145377

Academic Editor: Carlo Ricciardi

Received: 10 June 2022 Accepted: 12 July 2022 Published: 19 July 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). imposed a significant effect on the quality of people's lives. The development of healthcare systems requires a concerted effort to seamlessly integrate with the Internet of Things (IoT), especially for ameliorating day-to-day challenges arising in the sector. Recent developments in the health sector have consistently shown that combined technologies have the potential to improve healthcare services and assist healthcare professionals in the optimal and efficient delivery of healthcare solutions [1-3]. IoT is a new paradigm in technology which provides a conglomerate of novel services for the next wave of technological innovations [4]. IoT enables things (such as devices, cars, houses, people, and animals) to communicate with one another and with users over the internet network, thereby becoming an integral part of the Internet [5,6]. Furthermore, cloud computing services are used in IoT applications to create correct composite services by composing existing atomic services for IoT servicebased applications [7]. IoT applications also give users a lot of advantages, such as the ability to choose the best opportunity in each situation, to make decisions, to manage resources, and keep an eye on the environment's cloud resources [8]. A big part of the IoT is RFID, sensor technology, nanotechnology, and embedded intelligence technology. Each of the aforementioned technologies are being used to advance IoT applications for various purposes [2,9]. One of the major targets of the health sector is to realize high-quality healthcare delivery with low cost; the IoT has the power to make this real. For instance, the incorporation of sensor systems helps with better patient monitoring, leading to fewer tests, fewer unnecessary appointments and, consequently, lower costs. Hence, IoT technology is a key player in the early diagnosis and early intervention of diseases [10].

Significantly, the Internet of Medical Things (IoMTs) [11] or IoT (for the purposes of this paper, the terms IoT in healthcare and IoMT will be used interchangeably) will support the digital revolution, particularly in healthcare products. The developed IoMT applications prototype such as wearable devices enable patients, elderly people, or people with chronic diseases to remotely monitor their health status. In such cases, IoT applications can help in an emergency to quickly warn and alert caregivers or physicians of the elderly person [12].

Even though IoMT and its supporting technologies have been proven to mitigate health problems, such as medical errors, failure, ineffective workflows, and all evident benefits of IoT technologies in the healthcare sector, IoT systems are not fully integrated into healthcare organizations yet [13]. Additionally, IoT developments in the health sector have remained slow in terms of its implementation and adoption in other industries [14].

In light of the low adoption of IoT in the healthcare sector, it is difficult to implement it if users are not ready. Moreover, the decision to adopt IoT application requires a structured approach that is capable of identifying the technological and operational structures. Indeed, technology adoption is one of the mature areas of research in information systems, especially IoT adoption [15]. Huge investments are being made by companies and governments to adopt innovations that have the potential of bringing a paradigm shift in the user's lifestyle; for instance, the IoT technologies [16]. Several studies expect that healthcare professionals will have new responsibilities through using IoT [17]. Although the IoT can provide an improved and better approach to healthcare management, its end-user adoption is still very low especially among health-care professional staff [15].

Therefore, the objective of this work is to analyze and statistically classify the present research on the adoption of IoT among professional staff in healthcare, and to gain a comprehensive understanding of the adoption of IoT processes in healthcare. To meet the aforementioned objectives, several research questions (RQs) were formed. The novelty of this systematic literature review (SLR) provides the most recurrent adoption theories for IoT and the most recurrent factors that have a significant effect on its adoption in healthcare. This study classified IoT adoption factors into five categories which are related to individual factors, technology factors, security factors, health factors, and environmental factors. This paper provides an overview of the articles that are related to the adoption of IoT applications in healthcare from 2015 to 2021. To the best of our knowledge, there is no literature covering these points yet [18–22].

This study is more distinguished and unique as it explores new areas that previous studies rarely explore, or may have been referred to briefly. Furthermore, this study reviews the effective factors and criteria influencing IoT healthcare systems adoption. It used questionnaires, interviews, and expert opinions on the effective use of IoT. It also reviewed prior related research that provided new conceptual models related to the intention to use of IoT as a new technology. This SLR is constructed as follows: Section 2 provides an overview of IoT in healthcare. Section 3 elucidates the advantages of IoT in healthcare systems. The review strategy is provided in Section 4. Section 5 gives details on the characteristics of the included studies. Section 6 reviews IoT for testing and tracing. Section 7 discusses wearable devices used for IoT applications in healthcare. Section 8 includes regulations and procedures for IoT during pandemics. Section 9 discusses the findings of this SLR, which contains gaps and implications for future research, limitations, challenges of effective IoT's adoption, research directions, pervasive challenges across all verticals, healthcare during COVID-19 pandemic challenges, and data protection and privacy. Finally, Section 10 wraps up the paper with concluding remarks.

# 2. Overview of Deep Learning for IoT in Healthcare

IoT systems are made up of a large number of heterogeneous devices that are scattered across the network and generate a constant stream of data [23]. To design successful IoT applications, we often follow a workflow model composed of five components: formulation of the question, data gathering, data processing, visualization, and assessment. Even though obtaining hidden information and inferences from IoT data is a promising way to improve our lives, it is a difficult task that can not be done with traditional paradigms [24]. Indeed, with the birth of IoT devices, artificial intelligence (AI) has been introduced that uses continuous monitoring to help in illness diagnosis, notifying caregivers or physicians through an alert system. Apart from that, these gadgets may also be used to assist in decision-making through a decision support system (DSS). A significant advantage of this change was the transition of duties from a manual, stressful, and time-consuming process to a more intelligent, automated, and time-efficient one. Moreover, there have been occasions when medical practitioners have been unable to care for patients owing to a lack of information about emergency situations, resulting in disastrous choices, if not death. These machines are educated using specialized artificial intelligence algorithms, most often referred to as machine learning (ML) and deep learning (DL) algorithms [25]. DL has been actively utilized in many IoT applications in recent years [26]. It is a subset of machine learning (ML), and it is computationally intensive and costly. One of the issues is integrating DL techniques with IoT in order to increase the overall efficiency of IoT applications. Combining these strategies while maintaining a balance of computing cost and efficiency is critical for the next-generation of IoT networks [24].

Deep learning (DL) will be critical in developing a smarter IoT because it has demonstrated remarkable results in a variety of fields, including image recognition, information retrieval, speech recognition, natural language processing, indoor localization, and physiological and psychological state detection, amongst others. More also, these services serve as the foundation for IoT applications. Indeed DL has been actively utilized in many IoT applications in recent years [26,27].

#### 3. Advantages of IoT in Healthcare Systems

The following are the primary benefits or advantages of IoT technologies in healthcare systems that impact their adoption:

- 1. Cost savings: by being able to meet and assess patients remotely, the cost of in-person visits can be lowered [28–30]. Furthermore, with the introduction of home care devices, many patients can now be hospitalized and monitored at home.
- Treatment outcomes: because the monitoring is consistent, continuous, and automated, all data is kept in the cloud and provided to the doctor on a regular basis; the treatment processes were carried out correctly. The adoption of this strategy can

ensure that medical care is provided as soon as possible to examine the recovery process [31].

- 3. Disease management: by consistently recording and reporting a person's health indicators, diseases can be discovered and treated before they progress [32].
- 4. Error reduction: detailed and precise data collected automatically and free of human error can significantly reduce the rate of medical errors and their associated financial and critical costs [33,34].
- 5. Patient satisfaction: some factors such as the emphasis on the patient's requirements, data accuracy, timely treatment, cost reduction, reduction of repeated visits, recording of the recovery process, and, most importantly, the patient's active participation in the treatment process, have a positive impact on the patient [35].
- 6. Medication management: IoT assists patients in the precise use of drugs, as well as helping pharmacies and healthcare facilities in preventing drug waste [35,36].

## 4. Review Method

In this study, the SLR method was used, which is a way to find, analyze, and interpret all of the papers that have been written about a certain topic or research question [37]. The SLR process includes recognizing research, research questions, search strategy, study screening process and methods, quality evaluation, data extraction technique, and extracted data synthesis [37]. This systematic review was conducted for capturing relevant literature from different sources, focusing on the following objectives:

- 1. To explore conceptual frameworks for the adoption of IoT in healthcare.
- 2. To illustrate the future adoption of IoT in healthcare.

Overall, any SLR should be able to synthesize and analyze existing data on any subject, look for research gaps, and suggest the future direction on that subject [37]. Through investigating these objectives in detail, this review will make a significant enrichment in understanding the future adoption of IoT applications in the healthcare domain.

## 4.1. Information Sources

For this SLR, the articles were searched for in the following nine digital databases: Google Scholar, Science Direct, Emerald, Wiley, PubMed, IEEE, MDPI, Springer, and Scopus as represented in Figure 1.



Figure 1. Nine Digital Data Bases.

We carried out a search of the literature published between 2015 and 2021 related to IoT adoption among professional staff and IoT adoption in healthcare. Moreover, studies were selected from the databases using the following keyword combinations: Adoption, IoT, Nurses, (Adoption and IoMT and Nurses), (Adoption and IoT and Nursing care), (Adoption and IoT and Physician), (Adoption and IoMT and Physician), (Adoption and IoT in healthcare), and (Adoption and IoMT in healthcare).

## 4.2. Selection Criteria

The inclusion criteria for included review papers are:

1-Published between 2015 and 2021.

2-Written in English.

3-Available as full text.

4-Published in a peer-reviewed journal.

5-Articles and conference papers.

6-Articles investigating the adoption of IoT in healthcare.

The exclusion criteria are:

1-Duplicate studies.

2-Not written in English.

3-Not related in the adoption of IoT in healthcare.

4-Not available as full text.

5-Outside the time frame.

## 4.3. Research Question

Specifically, the goal of this part is to define the study questions in order to better understand the difficulties associated with the adoption of IoT in healthcare. The identification of the research question aids in the definition of the scope and goals of this study. The aims of this research are to investigate the area of IoT adoption in the healthcare sector and how the results of this study can be used to create a vision of the factors that assist the health industry in using IoT technology. To meet the research objective, the below four research questions (RQ) are formulated:

RQ1: What are the primary adoption areas that were selected by studies (adoption of IoT devices or end users' adoption for the chosen studies)?

RQ2: What theories/models were used in the studies?

RQ3: What constructs are being employed in the studies?

RQ4: Which kind of techniques are employed for data analysis in the chosen studies?

RQ5: What are the research gaps on in the current studies which are related to the use of IoT in the healthcare sector?

## 4.4. Quality Assessment

The quality assessment (QA) is defined as an accurate assessment about the general quality of the selected papers (N = 22). Several initiatives aimed at improving the overall quality of the search process were carried out throughout the assessment. Thus, to avoid any potential effects from previous searches, the online searches were conducted in incognito mode. The authors manually extracted relevant papers and articles from the first searches they conducted. In addition, when the abstracts were reviewed in detail, it was determined that certain publications had to be kept in the investigation and others needed to be dropped. In addition, a quality process for this SLR is developed by formulating five QA criteria as presented below:

- Q1: Is the paper's topic related to the adoption of IoT technology and its applications?
- Q2: Does the paper use adoption theories?
- Q3: Does the paper have theoretical framework and constructs?
- Q4: Does the paper explicitly present the research methodology?
- Q5: Is the procedure for collecting the research data clearly outlined in the paper?

A QA score is used to evaluate the study's overall quality as shown in Table 1. The study's overall quality is ranked on one of three quality levels: "high", "medium", and "poor" [38]. Research that entirely fulfills a quality requirement is given a score of 1 based on the resultant load score at the start of the investigation. Second, a score of 0.5 is provided to a candidate that just partly meets the requirements for consideration. After that, each item that obtains a score of 0 is awarded a score of 0 since it does not fulfill a quality standard, as defined by the five criteria for evaluation. With regard to the criteria, the greatest possible

loaded score for each research is 5, and the lowest possible loaded score is 0 for each study. (see Table 1). This article summarizes the findings of the QA process that was used on the 22 selected studies. The results show that 18 studies (82%) were classified as "high-quality", whereas 4 (18%) were classified as "medium-quality". There were no studies classified as "low-quality". As a consequence of this decision, no more research was omitted from the total. As a consequence, this SLR is mostly made up of 22 original papers that were carefully chosen.

PID	Q1	Q2	Q3	Q4	Q5	Scores
P1	1	1	1	0.5	1	4.5
P2	1	1	1	1	1	5
P3	1	1	1	1	1	5
P4	1	1	1	1	1	5
P5	1	1	1	1	1	5
p6	1	1	1	1	1	4
P7	1	1	0	0.5	1	3.5
P8	1	1	1	1	1	5
P9	1	1	1	1	1	5
P10	1	0	0	0	0	1
P11	1	1	1	1	1	5
P12	1	1	1	1	1	5
P13	1	1	1	1	1	5
P14	1	1	0	1	1	4
P15	1	1	0.5	1	1	4.5
P16	1	1	0	0	0	2
P17	1	1	1	1	1	5
P18	1	1	1	1	1	5
P19	1	1	1	1	1	5
P20	1	0	0	0	0	1
P21	1	1	1	1	1	5
P22	1	1	1	1	1	5

Table 1. Quality assessment scores for included review papers.

## 4.5. Results

Here, the prime concern was to find out the effective factors for the adoption of IoT in healthcare. The effective factors are those factors that affect the use of the healthcare system and also have an impact on user satisfaction [39]. Moreover, it also has to do with investigating the adoption of the IoT conceptual framework in healthcare, and adoption theories that are used in the adoption of IoT in the healthcare domain. The criteria for inclusion have been established as papers using the search keywords, which are mainly adoption, IoT in healthcare, and adoption of IoT by nurses, physicians, and patients.

Besides the title, abstract, and keywords of the papers that have been found, a list of notable articles has been made based on certain criteria. On the other hand, studies that focus on technical concerns and security challenges were left out of the analysis. The abstracts and titles of some of the studies were reviewed separately, to see if they met the criteria for being included. The procedure for search and selection of research material is illustrated in Figure 2.



Figure 2. Research Process.

# 4.5.1. Data Extraction and Organization

Based on the initial author's name and year of publication, as well as the kind of participants, the study design, and the geographical location of the research, if there is a theoretical framework, data were retrieved.

Table 2 shows the results of the quality evaluation for each of the selected studies and the elements extracted from each study. The authors extracted the data and double-checked it before publishing it. During the search approach, we reviewed all of the titles and abstracts of the selected papers. After that, we each looked at the full text of the articles that had been pre-selected and agreed on which one to choose. In addition, we did data extraction independently using a verified data extraction grid that was built based on the study criteria that we used. This generic data extraction grid has been designed and tested to categorize published publications according to existing theoretical frameworks, adoption theories, constructs, study locations, and design approaches. The data extraction grid in the Microsoft Excel program was filled in manually.

# 4.5.2. Included Studies

In the beginning, this research search strategy provided a total of 17,597 papers from databases such as PubMed, Wiley, Google Scholar, SCOPUS, Emerald, and Science Direct. The remaining 5327 papers were further screened after eliminating 1227 papers in the detection phase (unrelated to healthcare, technical papers, security issues, literature reviews, and duplicate studies). A total of 75 studies were considered worthy, but due to not meeting the inclusion criteria, 53 studies were excluded. Eventually, based on the

research goals, inclusion and exclusion requirements, 22 studies were included in this study and the full text of all the included studies has been retrieved.

Data Extraction	Description		
Study ID	A unique identifier		
Study Title	Title of each identified during the search.		
Author(s)	Author name.		
Year	Year of publication.		
Type of Participants	The type of user the paper conduct them.		
Research Design	Identification of the research methodology.		
Studies Place	Country/region where the research was undertaken.		
Theoretical Frameworks	Theory/model used by the selected papers.		
Adoption's Theory	Type of theory adoption used in the studies.		
Constructs	The constructs/factors used in the frameworks.		
Data collection strategy	Approach used to collect the data.		
Sample	Research participants.		
Type analysis and software	Software and type of analysis in the papers to obtain the result.		
Degree of article	A number indicating how much this study met the criteria for research quality.		

**Table 2.** Elements of the data extraction form with descriptions.

#### 5. Characteristics of Included Studies

In this review, we started conducting a search among publications between 2015 and 2021. However, we observed that in 2015 there were no publications related to our scope of study in all databases that we used. Figure 3 illustrates the studies' type. Few studies did not use theoretical frameworks. Figure 4 illustrates the number of articles growing exponentially from 2016 to 2021 [20,40]. In general, there are several theories, models, and frameworks that have been developed to explain user adoption of new technologies. The models introduce factors that can affect user acceptance such as the technology acceptance model (TAM), theory of planned behaviour (TPB), diffusion of innovation theory (DIT), theory of reasoned action (TRA), unified theory of acceptance and use of technology2 (UTAUT2), and Seddel model [41] (see Table 3).

Indeed, some of the selected studies for this SLR used other theories besides adoption theories to understand the effective factors in the adoption of IoT in healthcare, such as theories related to security and healthcare. For instance, cybernetic control theory (CCT), which is based on the idea of obtaining timely feedback, breaking down the deviations from the expectations, and taking important choices to address the deviations [42]. The theory of protection motivation (PMT) is concerned with the cognitive processes that mediate changes in attitudes and behaviors in healthcare [10]. Finally, the health belief model (HBM) is used "to describe and foresee health behaviors, to know the relationship of health behaviors, and utilize health services and practices systematically" [43].



Figure 3. Studies Type Included in Review.



Figure 4. Number of Articles Published by Year.

Table 3. The theories and their constructs that were used in each study.

Study	Adoption's Theory	Constructs		
[44]	(TRA) (TPB) (TAM)	Adoption intention Perceived behavioral control Perceived usefulness Perceived ease of use Subjective norm		
[40]	(UTAUT)	Performance Expectancy Effort expectancy Social InfluencePerceived RiskFacilitating ConditionsFinancial Cost Behavioral Intention		

Table 3. Cont.

Study	Adoption's Theory	Constructs
[45]	(UTAUT ) (UTAUT2)	Performance Expectancy Effort expectancy Social Influence Facilitating Conditions Prereceived creditability
[46]	(UTAUT)	Performance Expectancy Effort Expectancy Social Influence Perceived Risk Facilitating Conditions Perceived trust Behavioral Intention Age, gender, experience
[47]	(BRT)	Ubiquitous Reflective Reasons for Convenience Reflective Ubiquitous Reflective Relative advantage reflective Compatibility Reflective Reasons against Usage barrier reflective Risk barrier reflective Traditional barrier Reflective Attitude reflective Adoption intention reflective Value of openness to change
[10]	(TAM) (IDT) (PMT)	Perceived Advantage Technological Innovativeness Compatibility Trialability Image Perceived Vulnerability Perceived Severity Perceived Privacy Risk Cost Perceived Ease of Use Attitude
[48]	(DOI) (TAM)	Perceived usefulness Usefulness Necessariness Improvement Perceived usefulness
[49]	(UTAUT)	Performance Expectancy Effort expectation Behavior intention Behavioral to use
[14]	(DOI )	Perceived usefulness Perceived easy to use Computer self-efficiency Personal innovativeness Computer anxiety Services quality information quality
[50]	(FAHP)	Economic Prosperity Environmental Protection Quality of Life
[51]	(TAM) (TPB) (TRA) (UTAUT 2)	Trust organization Trust Provider trust treatment Trust technology
[52]	(TAM) (TPB) (TRA) (SE Theory)	Interpersonal influence, self-efficacy Attitude toward a wearable device Health interest, Perceived value trustworthiness

Study	Adoption's Theory	Constracts
[53]	(IDT) (TAM)	Perceived ease of use Behavioral intention cost Trialability compatibility attitude Privacy, Image self-efficiency Perceived usefulness
[54]	(UTAUT)	Not Mentioned
[42]	(CCT)	Information Pervasiveness Care Process Improvement M-IoT Adoption Care Service Efficiency
[55]	(TAM ) (UTAUT )	Performance expectancy
[43]	(HBM) (UTAUT)	Effort Expectancy Social Influence will Facilitating condition Performance expectancy Perceived severity Use behavior Trust Doctor's patient relation
[56]	(TAM)	Security Privacy, Trust in IoT Risk perception Familiarity Attitude
[57]	(TAM)	Behavioral Intention to Use Perceived Usefulness Perceived Ease of Use Attitude Perceived Connectedness Perceived Cost Privacy Concerns Perceived Convenience
[58]	Not Mentioned	Critical data management Unreliable results accuracy, security Unreliable results accuracy Unaffordable technology for low-income group lack of clear regulations. Critical data management Lack of clear regulations.
[59]	Saddon Model	Personal innovativeness E_loyalty Usefulness Personal innovativeness
[60]	(TAM) (HBM)	Perceived usefulness Consumer innovativeness Health information accuracy Reference group influence Health beliefs Privacy Protection

Table 3. Cont.

As Figure 5 depicts, the majority of studies used the TAM, UTAUT, and its updated 296 version UTATUT2 adoption theories [55]. However, TPB, TRA, DOI, BR theories [44], the IDT theory, HBM theory, Seddon model, CCT theory, IDM theory, PMI theory are used in fewer articles [42,44]. The studies about IoT adoption in healthcare were conducted in various countries including France, Spain, Germany, Sweden, Turkey, Hong Kong, and Israel. Developed and developing countries, such as India, Pakistan, Malaysia, Saudi Arabia, Iraq, Oman, and the Kingdom of Saudi Arabia from Asia and Latin America also recorded a large body of research in IoT adoption [43,53,57].



Adoption Theories Used in Publication

The majority of studies used quantitative research design (17.77%) [43,46]. One study employed a qualitative design through the use of focus groups (1.4%) [55], and representations of data collection methods. A mixed-method design was used in one study (1.4%) [46,54]. The rest of the studies did not mention the design methods they used [50]. The types of IoT applications studied were the following: a wearable device [55,61], general IoT devices (without specific any type) (9 papers), monitoring devices (2 papers) [52,58], rehabilitation device (1 paper), telehealth and behavior modification (1 paper) [42], smart city, and smart home (1 paper). Clinicians, nurses, medical workers, pharmacists, and other healthcare practitioners were involved in the included research (such as nutritionists, social workers, occupational therapists, and care services). Some other studies (4 papers) exclusively involved physicians [43,45,50,52]. The rest of the publications focused on patient (4 papers) [40,51,57,58], and end-users of IoT device in healthcare (9 papers) [40,52] (see Table 4).

Figure 5. Adoption theories used in publication.

Table 4. Data extraction for
------------------------------

SID	Study	Year	Type of Participants	Research Design	Studies Place	Theoretical Frameworks	Data Collection	Sample	Analysis and Software
S1	[44]	2017	Respondents in India	Not Mentioned	India	Yes	Survey	314	Partial Least Square SEM
S2	[40]	2020	Users of IoT-based healthcare devices	Quantitative Method	France	Yes	Survey	268	PLS-SEM
S3	[45]	2020	Younger physicians	Quantitative Method	Srilankan	Yes	Survey	375	SPSS
S4	[46]	2021	Patients	Quantitative Method	France	Yes	Online Survey	267	Partial Least Approach—Structural Equation Modeling
S5	[47]	2018	Older adults	Quantitative Method	Indian	Yes	Survey	815	PLS-SEM
S6	[10]	2018	End user IoT Product	Quantitative Method	Not Mentioned	Yes	Online Survey	426	SEM-PLS, and XLSTAT-PLSPM
S7	[48]	2020	The public user	Qualitative Method	Malaysia	No	Survey	Not Mentioned	Not Mentioned
S8	[49]	2020	Clinicians	Qualitative Method	Pakistan	Yes	Questionnaire	Over 479	PLS SEM
<b>S</b> 9	[14]	2020	Professionals or service administrators in healthcare	Mix Method	Saudi Arabia	Yes	Semi-Structured Interviews and Survey Data	Not Mentioned	NVIVO Software
S10	[50]	2018	applications	Not Mentioned	Not Mentioned	Yes	Not Mentioned	Not Mentioned	Fuzzy Logic
S11	[51]	2020	Patients	Quantitative Method	Not Mentioned	Yes	Questionnaire	117	PLS SEM
S12	[52]	2020	Device users	Quantitative Method	Germany and Sweden	Yes	Questionnaire	97	PLS SEM
S13	[53]	2020	Doctors	Quantitative Method	Iraq	Yes	Online Survey	250	SPSS
S14	[54]	2016	Physicians	Mixed-Methods	Israel	No	Questionnaire, Personal, and semi- Structured Interviews.	176	Microsoft Excel, and SPSS
S15	[42]	2019	Cardiologist Diabetologist Nutritionist	Quantitative Method	Not Mentioned	Yes	Online Survey	221	SEM
S16	[55]	2016	User wearable	Focus Group	Not Mentioned	No	Not Mentioned	Not Mentioned	Not Mentioned
S17	[43]	2017	Medical Doctors, Nursing Staff, and Patients	Quantitative Method	Pakistan	Yes	Survey	100	SPSS23
S18	[56]	2019	Users	Quantitative Method	Omani	Yes	Questionnaires	387	SPSS 25 and AMOS 25 statistics
S19	[57]	2019	Patient	Quantitative Method	Kingdom of Saudi Arabia	Yes	Survey	407	SEM
S20	[58]	2018	Patient	Quantitative Method	Latin-America,	No	Not Mentioned	Not Mentioned	Not Mentioned
S21	[59]	2018	Medical staff Care Services, Medical specialties, Covered Medical Facilities	Quantitative Method	Spain	Yes	Questionnaire	256	SPSS MEDIATE
S22	[60]	2019	Customers of Wearable Technology	Quantitative Method	Hong Kong	Yes	Online Survey	171	SmartPLS v3.28

#### 5.1. Overview of IoT Adoption Factors

In the results, 90 elements were identified as barriers or facilitators for IoT technology adoption and were classified in the different categories of factors from the extraction grid. As Figure 6 shows, these elements were classified as facilitators for IoT adoption in healthcare and barriers.

Indeed, a total of 61 elements (67%) are concerned with the category factors related to the IoT adoption in healthcare characteristics, in which 41 elements of them were identified as barriers and 20 elements as facilitators. The most repeated adoption factor was perceived usefulness [40,48,53]. Perceived usefulness is defined as an individual's perception that the utilization of IoT device will improve their performance of daily activities [53]. Perceived ease of use was another frequently mentioned factor [10]. Perceived ease of use is defined as an individual's perception that the utilization of IoT devices will be effortless [10,59].



Figure 6. Factors Characteristics.

As a result, it was critical for professionals to perceive the usefulness and ease of use of technology in their workplace; otherwise, there would be less incentive to use them. Other factors related to IoT in healthcare characteristics are cost issues, privacy, security concerns, and healthcare issues. The cost issues, privacy, security concerns, and healthcare were seen as barriers to the adoption of IoT in healthcare. Indeed, specialists were worried about the safety and confidentiality of the data contained in, and transmitted by, these technologies, as well as the possibility of device theft. Additionally, the cost of the IoT technology and its applications were perceived as barriers to IoT adoption in healthcare, and healthcare issues. The specialists are worried about the health risk if the technology is used in their health activities. Other factors identified in this category were individual factors, technology factors, security factors, health factors, and environmental factors.

## 5.2. Individual Factors

Individual factors in IoT adoption represented 16 elements (26%) among the total extracted elements. As shown in Figure 7 the most common factor identified was social influence factor, which is repeated six times [40,43,45,49,60]. Generally, professionals thought that IoT technology adoption is affected by social impact and conviction to receive support to use IoT technology in their activities, and it is used as facilitators more often than as barriers [40,45,49].

The next factor for individual adoption is attitude. It was mentioned five times [10,44,53,56,57]. Personal inattentiveness was mentioned four times as a facilitator [10,14,59,60], and self-efficacy, age, gender, and experience were mentioned three times each. The aforementioned factors are mentioned as facilitators more than as barriers [14,45,52]. Professionals believed that IoT adoption in healthcare could improve these factors. Finally, the next sets of factors are adoption intention, compatibility, and image. They were mentioned two times [10,52]. They were seen as barriers more than as facilitators. Furthermore, anxiety, interpersonal influence, value of openness to change, user satisfaction, e-loyalty, and perceived content-

edness were mentioned only one time. They were perceived as barriers [59]. Professionals believed that IoT adoption in healthcare could slow down due to these barriers.



**Individual Factors** 

#### Figure 7. Individual Factors.

#### 5.3. Technology Factors

IoT adoption technology factors represented 19 elements (31%) of the extracted elements. As shown in Figure 8 the most common factor identified was perceived usefulness factor, which was repeated six times [14,46,48,53,59,60]. As mentioned before, it was regarded as a facilitator. The professionals believed that the perceived usefulness factor in IoT adoption in healthcare is an essential factor as it improves their working conditions. The next factor related to IoT adoption in healthcare is perceived ease of use. It was repeated five times as a facilitator. The specialist is looking for technology that is easy to use during their duties. Thus, it is believed to be an important factor [14,48,53,57,59]. Performance expectancy factor is repeated four times [43,45,46,48], and effort expectation factor is repeated three times [14,43,46]. Both factors appeared as facilitators. Effort expectations are streamlined with the consensus based on the belief of the professionals that the use of IoT technology would be without effort. Performance expectancy is based on the belief that the use of the IoT technology would improve their performance [46], and from the professionals' perspective, these factors are considered as strong motivation to adopt IoT technology in their work. Additionally, trialability, and reasons for convenience reflective factors were repeated two times, [10,47]. Moreover, information quality, perceived advantage, technological innovativeness, perceived creditability, compatibility reflective, relative advantage reflec- tive, ubiquitous reflective, unreliable results accuracy, information pervasiveness, features deemed most useful, utilitarian hedonic benefits sought, features deemed most useful, and critical data management factors were repeated one time [10,47,55], and they mention it as barriers.



Figure 8. Technology Factors.

# 5.4. Security Factors

The main factors related to security for IoT adoption represented seven elements (11%) of the extracted elements. Perceived privacy risk, as shown in Figure 9 is the most common factor identified in the selected papers. It was repeated seven times, followed by trust which was repeated six times. In each instance, they were considered as facilitators [43,46,47,53,56,57,60]. Furthermore, perceived vulnerability, trust organization, trust provider, trust treatment, and trust technology are considered as barriers. They were repeated once each [51]. Indeed, the professionals believed that the security issue is a critical issue, especially in the case of using IoT technology in healthcare. This is because healthcare information needs a high level of privacy, and other professionals believe they could be threatened, misused or misinterpreted while using IoT technology.



Figure 9. Security Factors.

# 5.5. Health Factors

Health-related factors are essential to these studies. As shown in Figure 10, the most common factors identified in this category respectively are perceived severity and perceived health risk, where their repetition is two times [43,60]. Perceived severity "is explained as the degree of harm from unhealthy behavior" [43]. Furthermore, perceived health risk is defined as the degree to which the professional believes that using IoT will bring health problems [62]. This is followed by health interests, necessaries, health information accuracy, health beliefs, care process improvement, and care service efficiency, respectively; these appeared once each [52,55]. In this category, the studies consider these factors to be important, as they attempt to understand the adoption of IoT in health conditions. The professionals believed that using IoT technology in their duties could cause health problems related to the patient during the care process.



# **Health Factors**

Figure 10. Health Factors.

#### 5.6. Environment Factors

The last category encompasses external factors that exist inside the organizational environment and accounts for 7 elements (11%) of the extracted elements. Financial cost factor, as shown in Figure 11, is the most common factor, repeated four times [10,46,53,57]. The next facilitating conditions factor was repeated three times [43,45,46]. Other factors such as environmental protection, quality of life, traditional barrier reflectivity, lack of clear regulations, and technical infrastructure were spread equally at once each [47,50]. All environment factors are considered as barriers. The professionals believe that these environmental factors, especially both the cost factor and the facilitating conditions factor, affect the adoption of this technology. The higher the cost, the more the negative impact increased, particularly in long-term costs of the technology. The costs of the system and applications were mentioned in the studies. Moreover, if facilitating conditions are not applied to this development, the more negative impacts will be present in its adoption.



**Enviroment Factors** 

Figure 11. Environment Factors.

## 6. IoT for Testing and Tracing

COVID-19 testing and tracing enabled by the IoT can cut down the spread of the disease, which is very important in the fight against it. People are now more likely to use IoT to test and trace things, which is accelerating IoT adoption [63–65]. Thus, Figure 12 shows a test and trace system. Such advanced test and trace systems are composed primarily of two layers: data acquisition and data integration.



Figure 12. An Architecture of a Test and Trace IoT.

Data is gathered. Mobility data comes from a variety of sources, such as an immigration database, Global System for Mobile Communications (GSM)- and Global Positioning System (GPS)- enabled mobile phones, and Quick Response (QR) codes that can be tracked. This helps Taiwan keep track of both foreign and domestic travelers. Moreover, smart city resources are used, such as CCTV cameras to keep an eye on things [66]. Additionally, credit card transactions are monitored and recorded in order to discover consumers and forecast their behavior [67]. Individuals' hospitalizations are also logged and monitored to assist in locating missing persons [68].

A large number of countries and regions, including the United Kingdom, South Korea, Germany, Spain, Vietnam, and Taiwan, have adopted digital test-and-trace efforts [65,69]. While some of these attempts were unsuccessful in tracking the spread of COVID-19, nations that heavily used technology in their solutions did far better in their battle against

19 of 28

this pandemic. Ref. [70] reported that the United Kingdom's test and track system, which used around 27,000 contact tracers, had been unable to reach 21% of persons who tested positive for the week of 2–8 July 2020. Additionally, although 79% of those contacted identified 13,807 close connections, only 71% were contacted and urged to self-isolate.

Taiwan, on the other hand, had improved protocols in place (as a result of lessons learned from the 2003 SARS pandemic) and was able to suppress the initial COVID-19 wave. That is, as a result of swift and effective policy choices and widespread use of digital technology [71,72]. This indicates that a sophisticated technology-based test-and-trace system may be a viable tool for pandemic preparedness, provided all other safety measures are followed and rigorous policy choices are enforced. Figure 12 illustrates a structure for testing and trace system solutions.

A data integration layer combines and delivers data from disparate sources to the proper departments [73]. For example, the Centers for Disease Control and Prevention (CDCP) has access to the Immigration and National Health Insurance (NHI) databases [71]. Local governments are given access to location data. For example, surveillance data from a smart city setup is shared with police officers in order for them to take appropriate action [71]. Data extracted from various sources, such as public transportation and shopping malls, are also shared with the general public to assist them in making informed decisions about their daily routines [74].

# 7. Wearable Devices

While wearables such as smart watches, smart bands, and finger rings have long been available, the spread of COVID-19 has led to an upsurge in their appeal. When it comes to fighting COVID-19 and other future pandemics, wearable technology has a lot of promise [63,75,76]. Wearable technologies may also be used to broadcast health information [77]. Monitoring and connection tracing capabilities enforce social separation [78], provide tracking and contact tracing capabilities [79], enforce social separation [80], and provide mental healthcare [81] by continuously measuring an individual's cognition and mood, allowing individualized therapy interventions [82].

As a consequence of these and other comparable uses, wearable devices are becoming more popular. Smart wearables, according to Papa et al. (2020) [82], have the potential to transform healthcare. By 2024, the wearable industry is expected to reach USD 64 billion, according to Global Data [83,84]. A short discussion of recent successes in the battle against COVID-19 will follow.

A COVID-19 identification method has been created by WHOOP Inc. utilizing their WHOOP strap to assess respiratory rate using Resting Heart Rate (RHR). Using a mobile application, WHOOP strap data is sent to the WHOOP system [85]. Their technique detected 20% of COVID-19 positive people two days before the beginning of symptoms and 80% of COVID-19 positive persons by the third day of symptoms. Philips has also created temporary patches for the identification of COVID-19 patients in their early stages [86] and disposable biosensors for the early identification of deterioration in COVID-19 patients [87]. It measures and sends a number of indicators of deterioration, including respiratory rate, heartbeat, activity level, body position, and ambulation, among other things, to the doctor.

Artificial Intelligence (AI) can be used to accurately differentiate one disease from another [88,89]. Recently, AI was used by researchers from the Institute of Technology and Harvard University to determine whether COVID-19 subjects could be accurately differentiated from only a forced-cough cell phone recording [86]. Their findings, which were based on cough recordings from over 5000 subjects, show their model performed accurately. COVID-19 subjects were officially tested 97.1 percent of the time, and 100% of them were found to be asymptomatic when they were tested. Cough recordings have been used in the past to accurately diagnose circumstances such as pneumonia and asthma. This shows how useful it is to put these solutions into wearable devices so that they can provide a non-invasive, realistic solution for diagnosis of diseases, pre-screening, and outbreak monitoring [87,90]. A study by the Scrips Research Translations Institute is called Digital Engagement Tracking for Early Control Treatment (DETECT). DETECT gathered data from smartwatches and activity trackers that partners agreed to use, as well as self-reported symptoms and test results [91]. DETECT recently reported [92] that data from wearable devices can be used to identify COVID-19 cases with more accuracy than just symptoms alone. There have been several other studies such as this one [63,93,94]. This has accelerated deployment to allow interested individuals to voluntarily share their sensor and clinical data in order to combat COVID-19.

As a result of COVID-19, it has become increasingly critical for healthcare to be accessible, work on a low threshold, and be rapid in monitoring, testing and diagnosing [95]. Suddenly, we are faced with a brand-new set of issues. To begin, ubiquity implies that IoT-based medical services must be freely available and accessible to a greater population. However, the majority of people lack access to professional medical equipment, and must use less expensive technology such as smartphones and smart- watches, which have restricted healthcare capabilities. Second, since the threshold is low, the cost and difficulties of adopting IoT-based medical services must be reduced. On the other hand, medical care procedures are often complicated and expensive. Additionally, many individuals struggle to embrace and use new technology such as IoT solutions for healthcare. Third, it is necessary to address the energy needs and computational effectiveness of medical sensors in order to provide continuous tracking and high-quality testing/diagnosis [95].

#### 8. Regulations and Procedures for IoT during Pandemic

Previously, IoT adoption has been considerably lower in healthcare as a result of regulatory laws controlling privacy, data security, and approval processes. As a result of COVID- 19, emergency rules are being applied via established processes, and several innovative technologies have acquired emergency approvals driving IoT adoption. For instance, the United States Food and Drug Administration (USFDA, or FDA) has issued an Emergency Use Authorization for the electrocardiogram (ECG) low-ejection-fraction instrument created by Eko.

Eko is a company that makes digital health tools. They are based in the US and their tools assist health and clinicians providers in assessing cardiac complications associated with COVID-19 [96,97]. Eko makes it simpler for healthcare practitioners during ECG recordings to save, analyze, and communicate patient heart and lung sounds. Eko's machine learning algorithms were carefully developed utilizing different real-world data sets and are clinically proven to aid physicians in detecting early indicators of cardiac disease. The World Health Organization (WHO) has warned people about the risk of becoming sick in extremely crowded hospitals and emergency rooms. As a result of the regulation, the use of telehealth and home care has increased to cut down on hospital and clinic visits. Therefore, several countries are encouraging the use of telehealth services and have decided to add many types of medical services to their public health programs that can be accessed through telehealth. When people start using telehealth, they are using more IoT applications and other technologies. This is because the technology is becoming more common in the healthcare field. Ref. [98] shows how IoT technologies, smart telemedicine diagnosis systems, and virtual care work together for people of different ages and backgrounds [99,100].

# 9. Discussion

The whole issue of IoT technology in healthcare is gaining interest from companies and academics, since it provides a novel method of communicating with healthcare professionals and patients alike. Furthermore, it is a promising instrument to aid the healthcare industry [13]. The purpose of the study is to summarize the literature on factors that might support or prevent health professionals from using IoT technology in their job. Indeed, many nations own IoT devices and employ them in the healthcare sector. Nevertheless, this does not guarantee that the professionals adopt and accept it completely [15,101]. Therefore, it becomes important to study and identify the factors that may facilitate or impede healthcare professionals' use of IoT technologies. The review's primary results indicate that a variety of effective factors influence IoT adoption at the individual, technology, security, health, and environmental levels. The usefulness and ease of use of the technology were identified as two of the most critical factors influencing IoT adoption in the research. Furthermore, those two factors considered TAM theory factors. Moreover also, it is the most frequent theory in these studies for studying healthcare professional acceptance and adoption of IoT technology according to the literature. Moreover, most studies in this review are based on quantitative methods. The majority of the studies were conducted in developing countries [14,49,53]. In addition, our findings show that healthcare professionals think social influence factors in individual factors, privacy risk issues in security factors, perceived severity and perceived health risk in health factors, and cost issues in environment factors, could limit the adoption of the IoT technology. As mentioned, it has been noted in the literature that IoT technology may play a role in empowering patients and healthcare professionals [47,58]. In fact, medical experts feel that IoT technology in healthcare supports and improves doctor-patient relationships [49,58]. Additionally, the results show that healthcare providers agree that IoT in healthcare could improve patient care. We also assessed the results regarding prior studies conducted. We found that all studies considered the privacy risk issue in security factors as barriers [10,47,53]. Our findings in this review shows the same result. Additionally, some studies mentioned factors such as perceived severity and perceived health risk as having no significant effect on the adoption of IoT in healthcare [43]. Notwithstanding, other studies mentioned them as having an indirect effect on adoption [10]. Our analysis in this review found them to be regarded as barriers. Moreover, some studies consider individual factors to have no significant role in the adoption of technology [40], but the most studies are consider these to be facilitators [40,45]. Similarly, this review found the individual factors to be facilitators. Environmental factors in prior studies have been classified as barriers to adoption [55], and technology factors in all studies are considered facilitators [40,43]. This relationship is highlighted by our findings.

Finally, the results of this systematic review provide a baseline, allowing for a more comprehensive understanding of the challenges and opportunities associated with health-care professionals' utilization of IoT technology.

## 9.1. Gaps and Implications for Future Research

The majority of the studies included in the review consists of physicians as professionals in healthcare. In contrast, there are no studies conducted on nurses as professionals in healthcare even though nursing is considered as an essential segment for caring for patients in the healthcare sector. Moreover, nurses are considered the primary key potential users of IoT technology in healthcare. They play a large role in the adoption of IoT technology. Furthermore, the present studies used quantitative methods more than qualitative methods and mixed methods. Indeed, using different design methods provide methodological flexibility.

#### 9.2. Limitations

This study is not without limitations, which presents opportunities for further research. While this study provides an extensive compilation of current research about the factors influencing healthcare professional IoT adoption, it does have certain limitations. At the beginning, we searched the literature using just nine bibliographic databases, and we thoroughly evaluated the references of included studies as well. In addition, we include the papers citing those articles in order to reveal other possibly relevant publications. Secondly, in this study, we used a mixed-method systematic review approach to do a thorough review. It would have been better if we had used other methods, such as metanarrative or realism review, to acquire a more complete picture of how IoT is used in healthcare. People who do research in this area will find it very useful. Third, this review

only looked at data from studies that have been published. There was no extra contact with the authors to obtain more information or to make sure our classification was correct. Finally, we used a basic framework of adoption factors as the conceptual framework for categorizing elements identified as factors that affect IoT adoption in the studies included in this review and their repetitions in published studies. Even though we believe the framework is thorough and well-suited to introduce the adoption of IoT perceived by medical professionals, it is based on comprehensive, theoretical, and empirical research, but it may need further research.

However, future study may examine IoT adoption from a theoretical and empirical perspective.

#### 9.3. Challenges of Effective IoT's Adoption and Research Directions

COVID-19 has provided both possibilities and barriers to IoT adoption in the real world. Macroeconomically, IoT adoption must take into consideration the enormous cultural and economic shifts brought by COVID-19. Individual, community, and organizational behaviors have changed dramatically since the start of the global pandemic [102]. Prior to the pandemic, the worldwide IoT industry was developing. However, the accessibility and cost of device installation, and the security of data, have worsened. Microeconomically, IoT technology demands more quicker and revolutionary innovation in order to secure society's functioning, encourage civic building, and react to any future crises. Different IoT industries are experiencing new issues, and finding solutions that work will be crucial in speeding up IoT growth and acceptance in these areas [103].

#### 9.3.1. Pervasive Challenges across All Verticals (A) Financial Challenges

Numerous firms have cut or eliminated investment in a variety of new or planned projects, including IoT projects. Another financial concern exacerbated by COVID- 19 is the increased workers' cost for appliance installation in environments with social interaction limitations.

It is becoming more critical to design affordable, easy-to-install, and maintain IoT sensors and devices. For instance, it is vital to build low-power gadgets in order to save money on things such as battery replacement [104]. The Ultra Low Power SoC made by Dialog Semiconductor is a good example. It is used in battery-powered IoT devices [63], and has a battery life of more than a year in various IoT applications. Additionally, it is necessary to produce inexpensive plug-and-play sensing devices [92,105,106], as well as intelligent human-computer interaction [107], to simplify the installation and use of IoT systems.

#### 9.3.2. Data Protection and Privacy

During the COVID-19 pandemic, numerous countries have implemented a variety of emergency measures, including limits on mobility, social distance standards, and less rigorous privacy standards [108]. Following the pandemic's conclusion, such legislation will need to be thoroughly evaluated to guarantee that individual rights and privacy are respected [63]. Researchers need to build strong data access regulations, security standards, and privacy-preserving systems for tracking, monitoring, and analyzing, among other things, to be ready for new epidemics and emergencies [109]. Indeed, individual data pools on personal devices might be seen as a viable replacement for conventional data centers. They do not actively collect and upload data, but rather transmit it to consumer devices through produced data. It is possible for people to share data for data analysis and decision-making with people they trust or with third-party apps. Personal gadgets also need reliable encryption and network communication technologies. The system should smoothly incorporate 5G, edge computing, and blockchain [94,110].

# 9.3.3. Healthcare in COVID-19

COVID-19 is a healthcare catastrophe with obvious and immediate effects. According to a Juniper Research report, IoT platform revenue is predicted to expand by roughly 20% in 2020, from USD 55 billion in 2019 to USD 66 billion in 2020, and up from USD 35 billion in 2021. It has also found that revenues generated by IoT will exceed USD 4.3 billion by 2022–2023 [111], and according to [112] the annual economic impact of IoT in 2025 would be in the range of USD 2.7 to USD 6.2 trillion. Currently, three primary topics are driving IoT adoption in healthcare [111]

# 10. Conclusions

IoT has emerged as a new paradigm for improving healthcare. The health industry could potentially realize the advantages of IoT technologies as a result of the digital and information revolution [113,114]. The main benefits of the IoT are providing sustainable healthcare services, well-being, and more cost-effective treatment. The main goal of this study was to compile available literature related to IoT adoption and application for smart healthcare. This study examined, gathered, analyzed, and synthesized the essential data in a systematic manner. According to the results, the most commonly employed adoption theories for IoT adoption are TAM, TPB, TRA, and UTAUT. Furthermore, at the individual level, the key recognized adoption elements of IoT application in healthcare are social influence, attitude, and personal inattentiveness. Perceived usefulness, perceived ease of use, performance expectancy, and effort expectations are the IoT adoption factors at the technological level. Furthermore, perceived privacy risk is the most important issue at the security level. At the health level, the primary factors are perceived severity and perceived health risk, respectively; at the environmental level, the main factors are financial cost and facilitating conditions. The majority of the respondents participating in the included papers were physicians, patients, and healthcare professionals. Existing studies include various sorts of IoT applications such as wearable devices, monitoring devices, rehabilitation devices, telehealth and behavior modification, smart cities, and smart homes. In the year 2020, the majority of research on IoT adoption was done in France and Pakistan. This systematic review analyzes the critical characteristics that allow healthcare practitioners to recognize the constraints and opportunities for implementing IoT applications. Finally, we assess the anticipated impact of COVID-19 on IoT adoption in healthcare. After we thoroughly examined many of the studies in the literature, reports from prominent consulting companies, and interviews with specialists from a variety of sectors, we recognized the potential impact of COVID-19 on IoT adoption in various sectors, including healthcare, transportation, industrial IoT, smart homes, smart buildings, and smart cities, and how it pushed technology adoption and innovation. In addition, we address the different efforts that have been launched in a variety of areas in the aftermath of the epidemic. Furthermore, we discussed numerous obstacles that must be overcome as well as critical research paths that must be emphasized in order to expedite IoT adoption across the healthcare sector and other industries. Based on the review results, we can conclude that the main challenges and research directions for facilitating IoT adoption in various sectors are more related to wearables that are low-energy or harvest energy, wearables that are research-grade, AI algorithms for healthcare devices (for instance Artificial intelligence (AI)), and more accessible healthcare services. Furthermore, reducing the cost of developing, installing, and using IoT solutions and systems, as well as data security and privacy remains as concern in healthcare and other related industries. In general, adoption of the IoT is still limited to a few application areas. The results of this systematic study provide a common base for discussing the problems and opportunities associated with the adoption and usage of IoT by healthcare providers. Additionally, this study evaluated the work that has been done in this field, whether it be models and frameworks offered for enhancing or adopting IoT in healthcare, or proposed solutions to enable the realization of in real-world settings. However, capturing the full benefits of new technology while achieving a sustainable socioecological transition, still remains a challenge for government and the healthcare industry.

**Author Contributions:** Conceptualization, methodology, formal analysis M.A.-r.; writing—review and editing, writing—original draft preparation M.A.-r. and P.K.; supervision, project administration P.K. and B.B.; data curation, validation software A.Z. and M.A.; funding acquisition M.A., and P.K.; investigation, resources M.A. and A.Z. All authors have read and agreed to the published version of the manuscript.

**Funding:** This research is funded by the Office of Research & Sponsored Programs of Abu Dhabi University, United Arab Emirates, and the School of Computer Sciences, and Division of Research & Innovation, Universiti Sains Malaysia.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

**Data Availability Statement:** This article contains no data or material other than the articles used for the review and referenced.

**Conflicts of Interest:** The authors declare no conflict of interest.

## References

- Chiuchisan, I.; Costin, H.N.; Geman, O. Adopting the internet of things technologies in health care systems. In Proceedings of the 2014 International Conference and Exposition on Electrical and Power Engineering (EPE), Iasi, Romania, 16–18 October 2014; pp. 532–535. [CrossRef]
- 2. Mani, V.; Manickam, P.; Alotaibi, Y.; Alghamdi, S.; Khalaf, O.I. Hyperledger Healthchain: Patient-Centric IPFS-Based Storage of Health Records. *Electronics* 2021, *10*, 3003. [CrossRef]
- 3. Keikhosrokiani, P.; Mustaffa, N.; Zakaria, N. Success factors in developing iHeart as a patient-centric healthcare system: A multi-group analysis. *Telemat. Inform.* **2018**, *35*, 753–775. [CrossRef]
- 4. Ejaz, W.; Anpalagan, A. Internet of Things for Smart Cities: Technologies, Big Data and Security; Springer: Cham, Switzerland, 2019. [CrossRef]
- 5. Xu, D.L.; He, W.; Li, S. Internet of things in industries: A survey. IEEE Trans. Ind. Inform. 2014, 10, 2233–2243. [CrossRef]
- 6. Karthigaiveni, M.; Indrani, B. An efficient two-factor authentication scheme with key agreement for IoT based E-health care application using smart card. *J. Ambient. Intell. Humaniz. Comput.* **2019**, 1–12. [CrossRef]
- Selvaraj, S.; Sundaravaradhan, S. Challenges and opportunities in IoT healthcare systems: A systematic review. SN Appl. Sci. 2020, 2, 139. [CrossRef]
- 8. Adat, V.; Gupta, B. Security in Internet of Things: Issues, challenges, taxonomy, and architecture. *Telecommun. Syst.* 2018, 67, 423–441. [CrossRef]
- 9. Madakam, S.; Lake, V.; Lake, V.; Lake, V. Internet of Things (IoT): A literature review. J. Comput. Commun. 2015, 3, 164. [CrossRef]
- 10. Karahoca, A.; Karahoca, D.; Aksöz, M. Examining intention to adopt to internet of things in healthcare technology products. *Kybernetes* **2018**, *4*, 742–770. [CrossRef]
- 11. Keikhosrokiani, P.; Mustaffa, N.; Zakaria, N.; Baharudin, A.S. User behavioral intention toward using mobile healthcare system. In *Consumer-Driven Technologies in Healthcare: Breakthroughs in Research and Practice*; IGI Global: Hershey, PA, USA, 2019; pp. 429–444.
- 12. Manas, M.; Sinha, A.; Sharma, S.; Mahboob, M.R. A novel approach for IoT based wearable health monitoring and messaging system. *J. Ambient. Intell. Humaniz. Comput.* **2019**, *10*, 2817–2828. [CrossRef]
- 13. Mieronkoski, R.; Azimi, I.; Rahmani, A.M.; Aantaa, R.; Terävä, V.; Liljeberg, P.; Salanterä, S. The Internet of Things for basic nursing care A scoping review. *Int. J. Nurs. Stud.* 2017, *69*, 78–90. [CrossRef]
- Masmali, F.H.; Miah, S.J.; Mathkoor, N.Y. Internet of Things-based innovations in Saudi healthcare sector: A methodological approach for investigating adoption issues. In Proceedings of the 2020 IEEE Asia-Pacific Conference on Computer Science and Data Engineering (CSDE), Gold Coast, Australia, 16–18 December 2020; pp. 1–5. [CrossRef]
- 15. Bodur, G.; Gumus, S.; Gursoy, N.G. Perceptions of Turkish health professional students toward the effects of the internet of things (IOT) technology in the future. *Nurse Educ. Today* **2019**, *79*, 98–104. [CrossRef]
- 16. Straub, E.T. Understanding technology adoption: Theory and future directions for informal learning. *Rev. Educ. Res.* 2009, 79, 625–649. [CrossRef]
- 17. Islam, S.R.; Kwak, D.; Kabir, M.H.; Hossain, M.; Kwak, K.S. The internet of things for health care: A comprehensive survey. *IEEE Access* 2015, *3*, 678–708. [CrossRef]
- 18. Olushola, O.B. Factors affecting IoT adoption. IOSR J. Comput. Eng. (IOSR-JCE) 2019, 21, 19–24. [CrossRef]
- 19. Sujitparapitaya, S.; Shirani, A.; Roldan, M. Business intelligence adoption in academic administration: An empirical investigation. *Issues Inf. Syst.* **2012**, *13*, 112–122.
- Darshan, K.; Anandakumar, K. A comprehensive review on usage of Internet of Things (IoT) in healthcare system. In Proceedings of the 2015 International Conference on Emerging Research in Electronics, Computer Science and Technology (ICERECT), Mandya, India, 17–19 December 2015; pp. 132–136. [CrossRef]

- 21. Al-Turjman, F.; Nawaz, M.H.; Ulusar, U.D. Intelligence in the Internet of Medical Things era: A systematic review of current and future trends. *Comput. Commun.* **2020**, *150*, 644–660. [CrossRef]
- Ahmadi, H.; Arji, G.; Shahmoradi, L.; Safdari, R.; Nilashi, M.; Alizadeh, M. The application of internet of things in healthcare: A systematic literature review and classification. *Univers. Access Inf. Soc.* 2019, 18, 837–869. [CrossRef]
- Ma, X.; Yao, T.; Hu, M.; Dong, Y.; Liu, W.; Wang, F.; Liu, J. A survey on deep learning empowered IoT applications. *IEEE Access* 2019, 7, 181721–181732. [CrossRef]
- 24. Zikria, Y.B.; Afzal, M.K.; Kim, S.W.; Marin, A.; Guizani, M. Deep learning for intelligent IoT: Opportunities, Challenges and Solutions. *Comput. Commun.* **2020**, *164*, 50–53. [CrossRef]
- Durga, S.; Nag, R.; Daniel, E. Survey on machine learning and deep learning algorithms used in internet of things (IoT) healthcare. In Proceedings of the 2019 3rd International Conference on Computing Methodologies and Communication (ICCMC), Erode, India, 27–29 March 2019; pp. 1018–1022. [CrossRef]
- 26. Saleem, T.J.; Chishti, M.A. Deep learning for Internet of Things data analytics. Procedia Comput. Sci. 2019, 163, 381–390. [CrossRef]
- 27. Keikhosrokiani, P. IoT for enhanced decision-making in medical information systems: A systematic review. In *Enhanced Telemedicine and e-Health;* Springer: Cham, Switzerland, 2021; pp. 119–140.
- Yassein, M.B.; Hmeidi, I.; Al-Harbi, M.; Mrayan, L.; Mardini, W.; Khamayseh, Y. IoT-based healthcare systems: A survey. In Proceedings of the Proceedings of the Second International Conference on Data Science, E-Learning and Information Systems, Dubai, United Arab Emirates, 2–5 December 2019; Volume 30, pp. 1–9. [CrossRef]
- 29. Dauwed, M.; Meri, A. IOT Service Utilisation in Healthcare. In *Internet of Things (IoT) for Automated and Smart Applications;* Ismail, Y., Ed.; IntechOpen: Rijeka, Croatia, 2019; Chapter 4. [CrossRef]
- Mekki, K.; Bajic, E.; Chaxel, F.; Meyer, F. A comparative study of LPWAN technologies for large-scale IoT deployment. *ICT Express* 2019, 5, 1–7. [CrossRef]
- 31. Onasanya, A.; Elshakankiri, M. Smart integrated IoT healthcare system for cancer care. *Wirel. Netw.* **2021**, *27*, 4297–4312. [CrossRef]
- Kadhim, K.T.; Alsahlany, A.M.; Wadi, S.M.; Kadhum, H.T. An overview of patient 's health status monitoring system based on internet of things (IoT). Wirel. Pers. Commun. 2020, 114, 2235–2262. [CrossRef]
- 33. Mathew, P.S.; Pillai, A.S.; Palade, V. Applications of IoT in healthcare. In *Cognitive Computing for Big Data Systems Over IoT*; Springer: Cham, Switzerland, 2018; pp. 263–288. [CrossRef]
- Singh, R.P.; Javaid, M.; Haleem, A.; Suman, R. Internet of things (IoT) applications to fight against COVID-19 pandemic. *Diabetes Metab. Syndr. Clin. Res. Rev.* 2020, 14, 521–524. [CrossRef]
- 35. Kim, S.; Kim, S. User preference for an IoT healthcare application for lifestyle disease management. *Telecommun. Policy* **2018**, 42, 304–314. [CrossRef]
- Haleem, A.; Javaid, M.; Khan, I.H. Internet of things (IoT) applications in orthopaedics. J. Clin. Orthop. Trauma 2020, 11, S105–S106. [CrossRef]
- 37. Keele, S. *Guidelines for Performing Systematic Literature Reviews in Software Engineering*; Version 2.3 EBSE Technical Report; EBSE: Durham, UK, 2007. [CrossRef]
- Nidhra, S.; Yanamadala, M.; Afzal, W.; Torkar, R. Knowledge transfer challenges and mitigation strategies in global software development—A systematic literature review and industrial validation. *Int. J. Inf. Manag.* 2013, 33, 333–355. [CrossRef]
- Keikhosrokiani, P.; Mustaffa, N.; Zakaria, N.; Abdullah, R. Assessment of a medical information system: The mediating role of use and user satisfaction on the success of human interaction with the mobile healthcare system (iHeart). *Cogn. Technol. Work* 2020, 22, 281–305. [CrossRef]
- 40. Arfi, W.B.; Nasr, I.B.; Khvatova, T.; Zaied, Y.B. Understanding acceptance of eHealthcare by IoT natives and IoT immigrants: An integrated model of UTAUT, perceived risk, and financial cost. *Technol. Forecast. Soc. Chang.* **2021**, *163*, 120437. [CrossRef]
- Lai, P.C. The literature review of technology adoption models and theories for the novelty technology. J. Inf. Syst. Technol. Manag. 2017, 14, 21–38. [CrossRef]
- 42. Chakraborty, S.; Bhatt, V. Mobile IoT adoption as antecedent to Care-Service Efficiency and Improvement: Empirical study in Healthcare-context. *J. Int. Technol. Inf. Manag.* **2019**, *28*, 101–120.
- Solangi, Z.A.; Solangi, Y.A.; Aziz, M.S.A. An empirical study of Internet of Things (IoT) Based healthcare acceptance in Pakistan: PILOT study. In Proceedings of the 2017 IEEE 3rd International Conference on Engineering Technologies and Social Sciences (ICETSS), Bangkok, Thailand, 7–8 August 2017; pp. 1–7. [CrossRef]
- 44. Mital, M.; Chang, V.; Choudhary, P.; Papa, A.; Pani, A.K. Adoption of Internet of Things in India: A test of competing models using a structured equation modeling approach. *Technol. Forecast. Soc. Chang.* **2018**, *136*, 339–346. [CrossRef]
- Rajmohan, R.; Johar, M.G.M. Influence of social moderators on the Internet of things of Srilankan healthcare services. *Int. J. Recent Technol. Eng.* 2020, 9, 2154–2164. [CrossRef]
- 46. Arfi, W.B.; Nasr, I.B.; Kondrateva, G.; Hikkerova, L. The role of trust in intention to use the IoT in eHealth: Application of the modified UTAUT in a consumer context. *Technol. Forecast. Soc. Chang.* **2021**, *167*, 120688. [CrossRef]
- Sivathanu, B. Adoption of internet of things (IoT) based wearables for healthcare of older adults a behavioural reasoning theory (BRT) approach. *J. Enabling Technol.* 2018, 12, 169–185. [CrossRef]
- 48. Yuan, Y.; Cheah, T. A study of internet of things enabled healthcare acceptance in Malaysia. J. Crit. Rev. 2020, 7, 25–32. https://dx.doi.org/10.31838/jcr.07.03.04.

- 49. Dahri, A.S.; Thebo, L.A. An overview of AI enabled M-IoT wearable technology and its effects on the conduct of medical professionals in Public Healthcare in Pakistan. *3C Tecnol.* **2020**, *9*, 87. [CrossRef]
- Alansari, Z.; Anuar, N.B.; Kamsin, A.; Soomro, S.; Belgaum, M.R. The Internet of Things adoption in healthcare applications. In Proceedings of the 2017 IEEE 3rd International Conference on Engineering Technologies and Social Sciences (ICETSS), Bangkok, Thailand, 7–8 August 2017; pp. 1–5. [CrossRef]
- Bhatt, V.; Chakraborty, S. Importance of Trust in IoT based Wearable Device Adoption by Patient: An Empirical Investigation. In Proceedings of the 2020 Fourth International Conference on I-SMAC (IoT in Social, Mobile, Analytics and Cloud)(I-SMAC), Palladam, India, 7–9 October 2020; pp. 1226–1231. [CrossRef]
- 52. Hossain, M.I.; Yusof, A.F.; Hussin, A.R.C.; lahad, N.A.; Sadiq, A.S. Factors Influencing Adoption Model of Continuous Glucose Monitoring Devices for Internet of Things Healthcare. *Internet Things* **2021**, *15*, 100353. [CrossRef]
- Alhasan, A.; Audah, L.; Ibrahim, I.; Al-Sharaa, A.; Al-Ogaili, A.S.; Mohammed, J.M. A case-study to examine doctors' intentions to use IoT healthcare devices in Iraq during COVID-19 pandemic. *Int. J. Pervasive Comput. Commun.* 2020, *ahead-of-print.* [CrossRef]
- Biran Achituv, D.; Haiman, L. Physicians' attitudes toward the use of IoT medical devices as part of their practice. Online J. Appl. Knowl. Manag. 2016, 4, 128–145. [CrossRef]
- Canhoto, A.I.; Arp, S. Exploring the factors that support adoption and sustained use of health and fitness wearables. J. Mark. Manag. 2017, 33, 32–60. [CrossRef]
- 56. Alraja, M.N.; Farooque, M.M.J.; Khashab, B. The effect of security, privacy, familiarity, and trust on users' attitudes toward the use of the IoT-based healthcare: The mediation role of risk perception. *IEEE Access* **2019**, *7*, 111341–111354. [CrossRef]
- 57. Alanazi, M.; Soh, B. Behavioral intention to use IoT technology in healthcare settings. *Eng. Technol. Appl. Sci. Res.* 2019, 9,4769–4774. [CrossRef]
- Luna-delRisco, M.; Palacio, M.G.; Orozco, C.A.A.; Moncada, S.V.; Palacio, L.G.; Montealegre, J.J.Q.; Diaz-Forero, I. Adoption of Internet of Medical Things (IoMT) as an opportunity for improving public health in Latin America. In Proceedings of the 2018 13th Iberian Conference on Information Systems and Technologies (CISTI), Caceres, Spain, 13–16 June 2018; pp. 1–5. [CrossRef]
- 59. Martínez-Caro, E.; Cegarra-Navarro, J.G.; García-Pérez, A.; Fait, M. Healthcare service evolution towards the Internet of Things: An end-user perspective. *Technol. Forecast. Soc. Chang.* **2018**, *136*, 268–276. [CrossRef]
- Cheung, M.L.; Chau, K.Y.; Lam, M.H.S.; Tse, G.; Ho, K.Y.; Flint, S.W.; Broom, D.R.; Tso, E.K.H.; Lee, K.Y. Examining consumers' adoption of wearable healthcare technology: The role of health attributes. *Int. J. Environ. Res. Public Health* 2019, 16, 2257. [CrossRef]
- Walinjkar, A.; Woods, J. Personalized wearable systems for real-time ECG classification and healthcare interoperability: Real-time ECG classification and FHIR interoperability. In Proceedings of the 2017 Internet Technologies and Applications (ITA), Wrexham, UK, 12–15 September 2017; pp. 9–14. [CrossRef]
- Chaulagain, S.; Pizam, A.; Wang, Y. An integrated behavioral model for medical tourism: An American perspective. *J. Travel Res.* 2021, 60, 761–778. [CrossRef]
- 63. Umair, M.; Cheema, M.A.; Cheema, O.; Li, H.; Lu, H. Impact of COVID-19 on IoT adoption in healthcare, smart homes, smart buildings, smart cities, transportation and industrial IoT. *Sensors* 2021, *21*, 3838. [CrossRef]
- 64. Ribeiro-Navarrete, S.; Saura, J.R.; Palacios-Marqués, D. Towards a new era of mass data collection: Assessing pandemic surveillance technologies to preserve user privacy. *Technol. Forecast. Soc. Chang.* **2021**, *167*, 120681. [CrossRef]
- 65. Moon, J.; Chekar, C.K.; Barberá, D.; Davey, G.; Gaisser, S.; Gaisser, T.; Iwuji, C.; Meseguer, E.; Ryan, J.G.; Hopkins, M.M. Optimising 'Test and Trace' Systems: Early lessons from a comparative analysis of six countries. *SSRN* **2020**. [CrossRef]
- 66. Shorfuzzaman, M.; Hossain, M.S.; Alhamid, M.F. Towards the sustainable development of smart cities through mass video surveillance: A response to the COVID-19 pandemic. *Sustain. Cities Soc.* **2021**, *64*, 102582. [CrossRef]
- Chen, C.M.; Jyan, H.W.; Chien, S.C.; Jen, H.H.; Hsu, C.Y.; Lee, P.C.; Lee, C.F.; Yang, Y.T.; Chen, M.Y.; Chen, L.S.; et al. Containing COVID-19 among 627,386 persons in contact with the Diamond Princess cruise ship passengers who disembarked in Taiwan: Big data analytics. J. Med. Internet Res. 2020, 22, e19540. [CrossRef] [PubMed]
- Wang, C.J.; Ng, C.Y.; Brook, R.H. Response to COVID-19 in Taiwan: Big data analytics, new technology, and proactive testing. JAMA 2020, 323, 1341–1342. [CrossRef] [PubMed]
- 69. Cao, H.L.; Nguyen, H.A.D.; Luu, T.H.; Vu, H.T.T.; Pham, D.; Le, H.H.; Nguyen, D.X.B.; Truong, T.T.; Nguyen, H.D.; Nguyen, C.N.; et al. Localized automation solutions in response to the first wave of COVID-19: A story from Vietnam. *Int. J. Pervasive Comput. Commun.* **2020**. [CrossRef]
- 70. O 'Dowd, A. COVID-19: UK test and trace system still missing 80% target for reaching contacts. BMJ 2020, 370, m2875. [CrossRef]
- Lin, C.; Braund, W.E.; Auerbach, J.; Chou, J.H.; Teng, J.H.; Tu, P.; Mullen, J. Policy decisions and use of information technology to fight coronavirus disease, Taiwan. *Emerg. Infect. Dis.* 2020, 26, 1506. [CrossRef]
- Mathieu, E.; Ritchie, H.; Ortiz-Ospina, E.; Roser, M.; Hasell, J.; Appel, C.; Giattino, C.; Rodés-Guirao, L. A global database of COVID-19 vaccinations. *Nat. Hum. Behav.* 2021, 5, 947–953. [CrossRef]
- Su, S.F.; Han, Y.Y. How Taiwan, a non-WHO member, takes actions in response to COVID-19. J. Glob. Health 2020, 10, 010380. [CrossRef]
- 74. Zuo, F.; Wang, J.; Gao, J.; Ozbay, K.; Ban, X.J.; Shen, Y.; Yang, H.; Iyer, S. An interactive data visualization and analytics tool to evaluate mobility and sociability trends during COVID-19. *arXiv* 2020, arxiv:2006.14882.

- 75. Sawyer, J. Wearable Internet of Medical Things sensor devices, artificial intelligence-driven smart healthcare services, and personalized clinical care in COVID-19 telemedicine. *Am. J. Med. Res.* **2020**, *7*, 71–77. [CrossRef]
- 76. Maxwell, S.; Grupac, M. Virtual Care Technologies, Wearable Health Monitoring Sensors, and Internet of Medical Things-based Smart Disease Surveillance Systems in the Diagnosis and Treatment of COVID-19 Patients. Am. J. Med. Res. 2021, 8, 118–131. [CrossRef]
- 77. Seshadri, D.R.; Davies, E.V.; Harlow, E.R.; Hsu, J.J.; Knighton, S.C.; Walker, T.A.; Voos, J.E.; Drummond, C.K. Wearable sensors for COVID-19: A call to action to harness our digital infrastructure for remote patient monitoring and virtual assessments. *Front. Digit. Health* 2021, 2, 8. [CrossRef]
- Waheed, A.; Shafi, J. Successful role of smart technology to combat COVID-19. In Proceedings of the 2020 Fourth International Conference on I-SMAC (IoT in Social, Mobile, Analytics and Cloud)(I-SMAC), Palladam, India, 7–9 October 2020; pp. 772–777. [CrossRef]
- 79. Lin, L.; Hou, Z. Combat COVID-19 with artificial intelligence and big data. J. Travel Med. 2020, 27, 080. [CrossRef]
- 80. Hurley, D.; Popescu, G.H. Medical Big Data and Wearable Internet of Things Healthcare Systems in Remotely Monitoring and Caring for Confirmed or Suspected COVID-19 Patients. *Am. J. Med. Res.* **2021**, *8*, 78–90. [CrossRef]
- Vatansever, D.; Wang, S.; Sahakian, B.J. COVID-19 and promising solutions to combat symptoms of stress, anxiety and depression. *Neuropsychopharmacology* 2021, 46, 217. [CrossRef]
- 82. Papa, A.; Mital, M.; Pisano, P.; Del Giudice, M. E-health and wellbeing monitoring using smart healthcare devices: An empirical investigation. *Technol. Forecast. Soc. Chang.* 2020, 153, 119226. [CrossRef]
- 83. Miller, D.J.; Capodilupo, J.V.; Lastella, M.; Sargent, C.; Roach, G.D.; Lee, V.H.; Capodilupo, E.R. Analyzing changes in respiratory rate to predict the risk of COVID-19 infection. *PLoS ONE* 2020, *15*, e0243693. [CrossRef]
- Cohen, S.P.; Baber, Z.B.; Buvanendran, A.; McLean, B.C.; Chen, Y.; Hooten, W.M.; Laker, S.R.; Wasan, A.D.; Kennedy, D.J.; Sandbrink, F.; et al. Pain management best practices from multispecialty organizations during the COVID-19 pandemic and public health crises. *Pain Med.* 2020, *21*, 1331–1346. [CrossRef]
- 85. Philips Launches Next Generation Wearable Biosensor for Early Patient Deterioration Detection, Including Clinical Surveillance for COVID-19. Available online: https://www.philips.com/a-w/about/news/archive/standard/news/press/2020/202 00526-philips-launches-next-generation-wearable-biosensor-for-early-patient-deterioration-detection-including-clinical-surveillance-for-covid-19.html (accessed on 26 May 2020).
- 86. Laguarta, J.; Hueto, F.; Subirana, B. COVID-19 artificial intelligence diagnosis using only cough recordings. *IEEE Open J. Eng. Med. Biol.* **2020**, *1*, 275–281. [CrossRef]
- Porter, P.; Abeyratne, U.; Swarnkar, V.; Tan, J.; Ng, T.W.; Brisbane, J.M.; Speldewinde, D.; Choveaux, J.; Sharan, R.; Kosasih, K.; et al. A prospective multicentre study testing the diagnostic accuracy of an automated cough sound centred analytic system for the identification of common respiratory disorders in children. *Respir. Res.* 2019, 20, 1–10. [CrossRef]
- 88. Abiodun, O.I.; Jantan, A.; Omolara, A.E.; Dada, K.V.; Mohamed, N.A.; Arshad, H. State-of-the-art in artificial neural network applications: A survey. *Heliyon* **2018**, *4*, e00938. [CrossRef]
- Abiodun, O.I.; Jantan, A.; Omolara, A.E.; Dada, K.V.; Umar, A.M.; Linus, O.U.; Arshad, H.; Kazaure, A.A.; Gana, U.; Kiru, M.U. Comprehensive review of artificial neural network applications to pattern recognition. *IEEE Access* 2019, 7, 158820–158846. [CrossRef]
- 90. Gedam, S.; Paul, S. A review on mental stress detection using wearable sensors and machine learning techniques. *IEEE Access* **2021**, *9*, 84045–84066. [CrossRef]
- 91. Radin, J.M.; Wineinger, N.E.; Topol, E.J.; Steinhubl, S.R. Harnessing wearable device data to improve state-level real-time surveillance of influenza-like illness in the USA: A population-based study. *Lancet Digit. Health* **2020**, *2*, e85–e93. [CrossRef]
- Zhao, C.; Chen, K.Y.; Aumi, M.T.I.; Patel, S.; Reynolds, M.S. SideSwipe: Detecting in-air gestures around mobile devices using actual GSM signal. In Proceedings of the Proceedings of the 27th Annual ACM Symposium on User Interface Software and Technology, Honolulu, HI, USA, 5–8 October 2014; pp. 527–534. [CrossRef]
- 93. Quer, G.; Radin, J.M.; Gadaleta, M.; Baca-Motes, K.; Ariniello, L.; Ramos, E.; Kheterpal, V.; Topol, E.J.; Steinhubl, S.R. Wearable sensor data and self-reported symptoms for COVID-19 detection. *Nat. Med.* **2021**, *27*, 73–77. [CrossRef]
- 94. Omolara, A.E.; Alabdulatif, A.; Abiodun, O.I.; Alawida, M.; Alabdulatif, A.; Arshad, H. The internet of things security: A survey encompassing unexplored areas and new insights. *Comput. Secur.* 2022, *112*, 102494. [CrossRef]
- 95. Morrison, M.; Lăzăroiu, G. Cognitive Internet of Medical Things, Big Healthcare Data Analytics, and Artificial intelligence-based Diagnostic Algorithms during the COVID-19 Pandemic. *Am. J. Med. Res.* **2021**, *8*, 23–36. [CrossRef]
- 96. Abiodun, O.I.; Abiodun, E.O.; Alawida, M.; Alkhawaldeh, R.S.; Arshad, H. A review on the security of the internet of things: Challenges and solutions. *Wirel. Pers. Commun.* **2021**, *119*, 2603–2637. [CrossRef]
- 97. Dwivedi, R.; Mehrotra, D.; Chandra, S. Potential of Internet of Medical Things (IoMT) applications in building a smart healthcare system: A systematic review. *J. Oral Biol. Craniofacial Res.* 2022, *12*, 302–318. [CrossRef] [PubMed]
- Barr, J.R.; D 'Auria, D.; Persia, F. Telemedicine, Homecare in the Era of COVID-19 & beyond. In Proceedings of the 2020 Third International Conference on Artificial Intelligence for Industries (AI4I), Irvine, CA, USA, 21–23 September 2020; pp. 48–51. [CrossRef]

- Hutchings, O.R.; Dearing, C.; Jagers, D.; Shaw, M.J.; Raffan, F.; Jones, A.; Taggart, R.; Sinclair, T.; Anderson, T.; Ritchie, A.G. Virtual health care for community management of patients with COVID-19 in Australia: Observational cohort study. *J. Med. Internet Res.* 2021, 23, e21064. [CrossRef] [PubMed]
- Clarke, G. Remote Treatment of Patients during the COVID-19 Pandemic: Digital Technologies, Smart Telemedicine Diagnosis Systems, and Virtual Care. Am. J. Med. Res. 2020, 7, 29–35. [CrossRef]
- 101. Kang, S.; Baek, H.; Jung, E.; Hwang, H.; Yoo, S. Survey on the demand for adoption of Internet of Things (IoT) based services in hospitals: Investigation of nurses' perception in a tertiary university hospital. *Appl. Nurs. Res.* **2019**, *47*, 18–23. [CrossRef]
- 102. Ruiu, M.L. Mismanagement of COVID-19: Lessons learned from Italy. J. Risk Res. 2020, 23, 1007–1020. [CrossRef]
- 103. Irfan, M.; Ahmad, N. Internet of medical things: Architectural model, motivational factors and impediments. In Proceedings of the 2018 15th Learning and Technology Conference (L&T), Jeddah, Saudi Arabia, 25–26 February 2018; pp. 6–13. [CrossRef]
- 104. Gulati, M.; Parizi, F.S.; Whitmire, E.; Gupta, S.; Ram, S.S.; Singh, A.; Patel, S.N. CapHarvester: A stick-on capacitive energy harvester using stray electric field from AC power lines. *Proc. Acm Interact. Mob. Wearable Ubiquitous Technol.* 2018, 2, 1–20. [CrossRef]
- 105. Ravichandran, R.; Saba, E.; Chen, K.Y.; Goel, M.; Gupta, S.; Patel, S.N. WiBreathe: Estimating respiration rate using wireless signals in natural settings in the home. In Proceedings of the 2015 IEEE International Conference on Pervasive Computing and Communications (PerCom), St. Louis, MO, USA, 23–27 March 2015; pp. 131–139. [CrossRef]
- 106. Chatzigiannakis, I.; Hasemann, H.; Karnstedt, M.; Kleine, O.; Kröller, A.; Leggieri, M.; Pfisterer, D.; Römer, K.; Truong, C. True self-configuration for the IoT. In Proceedings of the 2012 3rd IEEE International Conference on the Internet of Things, Wuxi, China, 24–26 October 2012; pp. 9–15. [CrossRef]
- 107. Chhikara, P.; Singh, P.; Tekchandani, R.; Kumar, N.; Guizani, M. Federated learning meets human emotions: A decentralized framework for human–computer interaction for IoT applications. *IEEE Internet Things J.* **2020**, *8*, 6949–6962. [CrossRef]
- do Carmo Barriga, A.; Martins, A.F.; Simões, M.J.; Faustino, D. The COVID-19 pandemic: Yet another catalyst for governmental mass surveillance? *Soc. Sci. Humanit. Open* 2020, 2, 100096. [CrossRef]
- 109. Suganthi, S.; Anitha, R.; Sureshkumar, V.; Harish, S.; Agalya, S. End to end light weight mutual authentication scheme in IoT-based healthcare environment. *J. Reliab. Intell. Environ.* **2020**, *6*, 3–13. [CrossRef]
- 110. Fan, K.; Ren, Y.; Wang, Y.; Li, H.; Yang, Y. Blockchain-based efficient privacy preserving and data sharing scheme of content-centric network in 5G. *Iet Commun.* 2018, *12*, 527–532. [CrossRef]
- 111. Nasajpour, M.; Pouriyeh, S.; Parizi, R.M.; Dorodchi, M.; Valero, M.; Arabnia, H.R. Internet of Things for current COVID-19 and future pandemics: An exploratory study. *J. Healthc. Inform. Res.* **2020**, *4*, 325–364. [CrossRef]
- 112. Mohammadi, M.; Al-Fuqaha, A.; Sorour, S.; Guizani, M. Deep learning for IoT big data and streaming analytics: A survey. *IEEE Commun. Surv. Tutorials* **2018**, 20, 2923–2960. [CrossRef]
- 113. Keikhosrokiani, P. Predicating smartphone users' behaviour towards a location-aware IoMT-based information system: An empirical study. *Int. J. E-Adopt.* **2021**, *13*, 52–77. [CrossRef]
- 114. Keikhosrokiani, P. Perspectives in the Development of Mobile Medical Information Systems: Life Cycle, Management, Methodological Approach and Application; Academic Press: Cambridge, MA, USA, 2019. [CrossRef]